Author's response to reviews

Title: Are osteoporotic fractures being adequately investigated?: A questionnaire of GP & orthopaedic surgeons

Authors:

George Chami (georgechami@doctors.org.uk)
Lee Jeys (lee.jeys@btclick.com)
Mathew Freudmann (m@freudmann.co.uk)
Louise Connor (louise_connor@hotmail.com)
Mashood Siddiqi (masiddiqi@doctors.org.uk)

Version: 6  Date: 21 January 2006

Author's response to reviews:

Dear Sir or Madam,

Thank you for your email,

please note that the following changes has been made to the manuscript:

1- The references has been reformatted using End Note.


3 In page 3 the title of the paragraph "Method" changed to "Methods"

Kind Regards

George Chami